

# **JGR** Biogeosciences

# **RESEARCH ARTICLE**

10.1029/2021JG006690

#### **Key Points:**

- We found evidence for lags in allocation of previous year's carbon uptake to wood biomass increment (WBI) at three of six sites studied
- Sites with high annual WBI demonstrated strong correlations to carbon uptake, with the strongest correlations including previous year uptake
- WBI was less tightly coupled to carbon uptake in less productive forests that also exhibited low interannual variability in WBI

#### **Supporting Information:**

Supporting Information may be found in the online version of this article.

Correspondence to:

A. Teets, aft49@nau.edu

#### Citation:

Teets, A., Moore, D. J. P., Alexander, M. R., Blanken, P. D., Bohrer, G., Burns, S. P., et al. (2022). Coupling of tree growth and photosynthetic carbon uptake across six North American forests. *Journal of Geophysical Research: Biogeosciences*, *127*, e2021JG006690. https://doi. org/10.1029/2021JG006690

Received 28 OCT 2021 Accepted 14 MAR 2022

#### **Author Contributions:**

**Conceptualization:** Aaron Teets, David J. P. Moore, Mariah S. Carbone, Shawn Fraver, Christopher M. Gough, David Y. Hollinger, Neil Pederson, Andrew D. Richardson

Data curation: Aaron Teets, David J. P. Moore, M. Ross Alexander, Peter D. Blanken, Gil Bohrer, Sean P. Burns, Christopher M. Gough, David Y. Hollinger, J. William Munger, Kimberly A. Novick, Scott V. Ollinger, Andrew P. Ouimette, Neil Pederson, Christoph S. Vogel, Andrew D. Richardson Formal analysis: Aaron Teets Funding acquisition: Andrew D. Richardson Investigation: Aaron Teets, Andrew D. Richardson

© 2022. American Geophysical Union. All Rights Reserved.

# Coupling of Tree Growth and Photosynthetic Carbon Uptake Across Six North American Forests

Aaron Teets<sup>1,2</sup> , David J. P. Moore<sup>3</sup> , M. Ross Alexander<sup>4,5</sup> , Peter D. Blanken<sup>6</sup> , Gil Bohrer<sup>7</sup> , Sean P. Burns<sup>6,8</sup> , Mariah S. Carbone<sup>1,2</sup> , Mark J. Ducey<sup>9</sup> , Shawn Fraver<sup>10</sup> , Christopher M. Gough<sup>11</sup> , David Y. Hollinger<sup>12</sup> , George Koch<sup>1,2</sup> , Thomas Kolb<sup>13</sup> , J. William Munger<sup>14</sup> , Kimberly A. Novick<sup>15</sup> , Scott V. Ollinger<sup>16</sup> , Andrew P. Ouimette<sup>16</sup> , Neil Pederson<sup>4</sup> , Daniel M. Ricciuto<sup>17</sup> , Bijan Seyednasrollah<sup>18</sup> , Christoph S. Vogel<sup>19</sup> , and Andrew D. Richardson<sup>1,18</sup>

<sup>1</sup>Center for Ecosystem Science and Society, Northern Arizona University, Flagstaff, AZ, USA, <sup>2</sup>Department of Biological Sciences, Northern Arizona University, Flagstaff, AZ, USA, <sup>3</sup>School of Natural Resources and the Environment, University of Arizona, Tucson, AZ, USA, <sup>4</sup>Harvard Forest, Harvard University, Petersham, MA, USA, <sup>5</sup>Midwest Dendro, LLC, Naperville, IL, USA, <sup>6</sup>Department of Geography, University of Colorado, Boulder, CO, USA, <sup>7</sup>Department of Civil, Environmental and Geodetic Engineering, Ohio State University, Columbus, OH, USA, <sup>8</sup>National Center for Atmospheric Research, Boulder, CO, USA, <sup>9</sup>Department of Natural Resources and the Environment, University of New Hampshire, Durham, NH, USA, <sup>10</sup>School of Forest Resources, University of Maine, Orono, ME, USA, <sup>11</sup>Department of Biology, Virginia Commonwealth University, Richmond, VA, USA, <sup>12</sup>USDA Forest Service, Northern Research Station, Durham, NH, USA, <sup>13</sup>School of Forestry, Northern Arizona University, Flagstaff, AZ, USA, <sup>14</sup>School of Engineering and Applied Sciences, Harvard University, Cambridge, MA, USA, <sup>15</sup>O'Neill School of Public and Environmental Affairs, Indiana University - Bloomington, Bloomington, IN, USA, <sup>16</sup>Earth Systems Research Center, University of New Hampshire, Durham, NH, USA, <sup>17</sup>Oak Ridge National Laboratory, Environmental Sciences Division and Climate Change Science Institute, Oak Ridge, TN, USA, <sup>18</sup>School of Informatics, Computing, and Cyber Systems, Northern Arizona University, Flagstaff, AZ, USA, <sup>19</sup>University of Michigan Biological Station, Pellston, MI, USA

**Abstract** Linking biometric measurements of stand-level biomass growth to tower-based measurements of carbon uptake—gross primary productivity and net ecosystem productivity—has been the focus of numerous ecosystem-level studies aimed to better understand the factors regulating carbon allocation to slow-turnover wood biomass pools. However, few of these studies have investigated the importance of previous year uptake to growth. We tested the relationship between wood biomass increment (WBI) and different temporal periods of carbon uptake from the current and previous years to investigate the potential lagged allocation of fixed carbon to growth among six mature, temperate forests. We found WBI was strongly correlated to carbon uptake across space (i.e., long-term averages at the different sites) but on annual timescales, WBI was much less related to carbon uptake, suggesting a temporal mismatch between C fixation and allocation to biomass. We detected lags in allocation of the previous year's carbon uptake, with the strongest correlations to carbon uptake from the previous year's carbon uptake, with the strongest correlations to carbon uptake from the previous year. Forests with low rates of WBI demonstrated weak correlations to carbon uptake from the previous year. Forests with low rates of WBI demonstrated weak correlations to carbon uptake from the previous year and stronger relationships to current year climate conditions. Our work shows an important, but not universal, role of lagged allocation of the previous year's carbon uptake to growth in temperate forests.

**Plain Language Summary** We compared the interannual variability of stand-level biomass growth (estimated from annual tree-level measurements) to carbon uptake (measured from towers monitoring gas exchange over forest canopies) to identify temporal mismatches between the two processes. We used data from multiple forested sites with long-term measurements of carbon uptake to ask the question: is there a consistent temporal offset between the uptake of carbon and the allocation to plant biomass? We found that the relationship between growth and carbon uptake varies among sites, and there was no consistent temporal offset between the uptake of carbon and allocation to biomass growth. Sites with higher growth rates had higher interannual variability, and more apparent coupling between biomass growth and carbon uptake from the previous year. Forests with lower growth rates had weaker relationships with carbon uptake and stronger coupling with current year environmental conditions. We demonstrate that temporal lags between carbon uptake and allocation to growth are not universal among these temperate forests, and the carry-over of uptake stored from the previous year is not as critical in slow-growing forests, compared to fast-growing forests, likely due



Methodology: Aaron Teets, David J. P. Moore, Mark J. Ducey, Shawn Fraver, George Koch, Thomas Kolb, Daniel M. Ricciuto, Bijan Seyednasrollah, Andrew D. Richardson

**Project Administration:** Andrew D. Richardson

Resources: Andrew D. Richardson Supervision: David J. P. Moore, Mariah S. Carbone, Andrew D. Richardson Visualization: Aaron Teets, Andrew D. Richardson

Writing – original draft: Aaron Teets, Andrew D. Richardson Writing – review & editing: Aaron Teets, David J. P. Moore, Peter D. Blanken, Gil Bohrer, Sean P. Burns, Mariah S. Carbone, Mark J. Ducey, Shawn Fraver, Christopher M. Gough, David Y. Hollinger, George Koch, Thomas Kolb, J. William Munger, Kimberly A. Novick, Scott V. Ollinger, Andrew P. Ouimette, Neil Pederson, Bijan Seyednasrollah, Andrew D. Richardson to lower demands for growth. This work helps to clarify the limitations on the growth of slow-turnover wood biomass.

### 1. Introduction

Biomass stored in trees represents the largest pool of living carbon on the planet (Houghton et al., 2009), and the growth of new tree tissues is one of the largest and most persistent global sinks of atmospheric carbon (Pan et al., 2011). Once carbon is allocated to structural growth in trees, as lignin or cellulose, it becomes relatively recalcitrant (Alexander et al., 2018; Russell et al., 2014) and provides the foundation for long-term carbon seques-tration in forests. Direct measurements of tree growth are commonly used to estimate the rates of carbon moving into forest ecosystems. Tree rings (Babst et al., 2014; Bouriaud et al., 2005) or repeated annual biometric inventories (Clark et al., 2001) combined with allometric equations are used to quantify whole forest wood biomass increment (WBI) with annual resolution. We refer to WBI as the stand-level mass of carbon allocated to boles, branches, and coarse roots of trees annually. Long-time series of WBI are useful for several different applications, including carbon accounting purposes (e.g., Hollinger et al., 2021), reconciling ecosystem models (Xu et al., 2017), and monitoring forest carbon accumulation in forest ecosystems through time.

Despite the importance of wood biomass production to the global carbon cycle, the environmental controls on the interannual variability of WBI remain only partially understood. This is largely because we do not understand how WBI is related to the source of carbon needed for growth—photosynthetic carbon uptake (Sala et al., 2012). At the ecosystem level, metrics of carbon uptake (i.e., net ecosystem productivity [NEP] and gross primary productivity [GPP]) are measured from towers using high-frequency observations of the covariance between the vertical wind speed and atmospheric CO<sub>2</sub> concentration (Baldocchi et al., 1988). While multiyear averages of WBI are considered highly correlated with multiyear averages of NEP (Pregitzer & Euskirchen, 2004) and GPP (Waring et al., 1998; Zha et al., 2013), the interannual variability of these metrics are oftentimes decoupled from one another (e.g., Delpierre et al., 2016). Understanding the relationship between WBI and carbon uptake is a critical step in understanding the environmental controls on WBI.

In the last two decades, researchers have tested the relationship between WBI and annual carbon uptake (measured from eddy covariance towers) across several different forest types to better understand the limitations on stand-level biomass growth (e.g., Babst et al., 2013). While some studies have found strong relationships between the interannual variability of WBI and annual carbon uptake (e.g., Lempereur et al., 2015; Ohtsuka et al., 2009), others have found carbon uptake over calendar-year timescales to be largely decoupled from WBI (e.g., Barford et al., 2001; Curtis et al., 2002; Teets et al., 2018). Some studies found subannual integrals of carbon uptake are better predictors of WBI (Babst et al., 2013; Gea-Izquierdo et al., 2014; Granier et al., 2008) because they have the benefit of excluding winter seasons when temperate trees are not actively photosynthesizing. Despite the growing amount literature on this topic (also see Anić et al., 2018; Guillemot et al., 2015; Rocha et al., 2006), it is difficult to synthesize this information due to (a) different metrics and analytical approaches used to characterize relationships between WBI and carbon uptake, (b) different forest types studied, and (c) varying climatology affecting trees where the studies are located. In addition, these studies rarely test the relationship between WBI and carbon uptake from the previous year.

It is possible that weak correlations found between stand-level growth and carbon uptake in many of these studies are due to delays in carbon allocation (i.e., lags) until growth the following year (see Gough et al., 2009; Richardson et al., 2013). Lags in allocation are commonly associated with negative conditions for WBI—when stressful conditions (e.g., droughts) can cause negative growth effects lasting for several years (Anderegg et al., 2015; Kannenberg et al., 2020; Ogle et al., 2015). But lags in allocation of carbon to structural growth can also occur under normal environmental conditions. When carbon uptake exceeds the demand for growth surplus photosynthate is allocated to storage as nonstructural carbohydrates (Hartmann et al., 2015; Kozlowski, 1992). Allocation of previously stored nonstructural carbohydrates to growth in following years has been demonstrated using isotopic tracers (e.g., Epron et al., 2012; Keel et al., 2006; Mildner et al., 2014) and delayed allocation of carbohydrates is thought to be largely responsible for autocorrelation found in tree-ring widths (Carbone et al., 2013; Kagawa et al., 2006). Although delayed responses of past year's conditions on tree growth has been recognized by dendrochronologists for several decades (e.g., Fritts, 1976), we still do not understand the mechanisms explaining different lagged growth responses between species (Zweifel & Sterck, 2018) and forest types.





**Figure 1.** Conceptual diagram of correlation matrices used in this study. Each  $1 \times 1$  quadrant represents a variable integrated over a different temporal period (start of integral on *x*-axis, and end of integral on *y*-axis). The x- and *y*-axes origin represents January of the year before growth (the prefix *p* refers to previous years). The 1-to-1 line (solid line) represents 1-month integrals, the shortest temporal period used for this study. The dashed line represents 12-month integrals (the area above the line represents temporal periods >1 year, and the area below the line is less than 1 year). The dotted lines separate the two calendar years. The lower left quadrant represents only previous year integrals, the top right quadrant current year integrals, and the top left including portions of both current and previous year metrics. Shape outlines refer to different seasons: previous year summer ( $\bigcirc$ ), previous year autumn ( $\square$ ), spring ( $\diamondsuit$ ), and summer ( $\triangle$ ) uptake.

Lagged allocation of carbon can also occur due to seasonal differences between growth and photosynthesis. Compared to photosynthesis, growth of new wood tissues in cold-climate temperate forests occurs over relatively short temporal periods in the late spring and summer (D'Orangeville et al., 2021; Rossi et al., 2006) because xylogenesis (i.e., wood formation) is more sensitive to low temperatures (Körner, 1998) and moisture deficits (Estiarte & Penuelas, 1999; Kozlowski & Pallardy, 2002) than is photosynthesis. In temperate forests, reserves of sugars and starch (i.e., nonstructural carbohydrates) accumulate once growth has completed and are typically highest just before winter dormancy (Furze et al., 2018). At times, it is thought that growth is actively downregulated to allow the accumulation of nonstructural carbohydrates (Dietze et al., 2014). When favorable conditions return, trees can remobilize stored carbohydrates from the previous year to initiate growth in the spring (Eglin et al., 2010; Kuptz et al., 2011). Documenting lagged allocation of carbon stand-level growth would validate the importance of lagged growth responses at the ecosystem scale and has rarely been tested (but see Gough et al., 2009).

In this study, we test whether temporal lags in carbon allocation to stand-level growth are found across six different North American forests. We ask, is annual WBI better explained by carbon uptake (i.e., NEP and GPP) from the previous year (i.e., lags in allocation) compared to current year carbon uptake? We explore the research question systematically at multiple sites to test whether the temporal relationship between WBI and carbon uptake is conserved or varies among forests. Our null hypothesis was that WBI exhibits significant positive correlations to only current year carbon uptake among forests, suggesting little reliance on previously stored carbohydrates for growth. Our alternate hypothesis was that WBI, across all sites studied, exhibits strong positive correlations with carbon uptake integrals from the previous year, suggesting there are consistent and measurable lags in the allocation of stored nonstructural carbohydrates for later growth. To test these hypotheses, we identified the periods of carbon uptake most strongly correlated to WBI using a correlation matrix approach (as used in Babst et al., 2013) to characterize the relationships between WBI and different temporal periods of carbon uptake at the different sites (see conceptual figure of the hypotheses in Figure 1). Following, we tested whether WBI responds to current or previous year climate drivers using the same methodology. By identifying the drivers of WBI, we can better understand the factors that regulate the stand-level growth of long-lived wood biomass, an important process that



#### Table 1

Site and Data Description for AmeriFlux Sites Listed by Increasing Mean Annual Temperature (and Decreasing Conifer Cover) From Left to Right

	Niwot Ridge	Howland Forest	Bartlett Experimental Forest	University of Michigan Biological Station	Harvard Forest	Morgan Monroe State Forest
Site ID	US-NR1	US-Ho1	US-Bar	US-UMB	US-Ha1	US-MMS
Location	Colorado, U.S.	Maine, U.S.	New Hampshire, U.S.	Michigan, U.S.	Massachusetts, U.S.	Indiana, U.S.
Mean annual temperature (°C)	1.5	5.3	5.6	5.8	6.6	10.9
Mean annual precipitation (mm)	800	1,070	1,246	803	1,071	1,032
Elevation (m)	3,050	60	272	234	340	275
Dominant genera in decreasing order	Abies, Pinus, and Picea	Picea and Tsuga	Acer, Fagus, and Betula	Populus, Quercus, and Pinus	Quercus and Acer	Liriodendron, Acer, and Quercus
EPA <sup>a</sup> Ecoregion Level III	Southern Rockies	Acadian Plains and Hills	Northeastern Highlands	Northern Lakes and Forests	Northeastern Highlands	Eastern Temperate Forests
QMD <sup>b</sup> (cm) in 2012	19	29	26	24	37	44
Living tree biomass <sup>c</sup> in 2012 (Mg C $ha^{-1} \pm SD$ )	46 ± 11	138 ± 18	96 ± 16	64 ± 12	153 ± 34	121 ± 39
Leaf area index $(m^2 m^{-2})^d$	3.8–4.2	5.3-6.3	4.0–5.5	3.4–4.4	4.5–5.5	4.2–5.2
References	Burns et al. (2015); Monson et al. (2005)	Hollinger et al. (2004)	Ouimette et al. (2018)	Gough et al. (2013)	Urbanski et al. (2007)	Schmid et al. (2000)
Data set DOI	Blanken et al. (1998; 10.17190/ AMF/12460880)	Hollinger (1996; 10.17190/ AMF/1246061)	Richardson (2004; 10.17190/ AMF/1246030)	Gough et al. (1999; 10.17190/ AMF/1246107)	Munger (1991; 10.17190/ AMF/1246059)	Novick and Phillips (1999; 10.17190/ AMF/1246080)

 $^{a}$ EPA = U.S. Environmental Protection Agency.  $^{b}$ QMD = quadratic mean diameter of the trees sampled.  $^{c}$ Uncertainty calculated from Monte Carlo sampling with replacement of individual tree biomass.  $^{d}$ LAI was estimated from harvesting trees and scaling to the stand level at US-NR1(Turnipseed et al., 2002), litter collection at US-UMB, and using the LAI 2000 plant canopy analyzer (Li-Cor Inc, Lincoln, NE) at US-Ho1, US-Bar, US-Ha1, and US-MMS.

remains only partially understood (Dietze et al., 2014). Characterizing the link between WBI and carbon uptake is therefore an important step in understanding forest carbon cycling.

# 2. Materials and Methods

#### 2.1. Description of Data Sources

We compiled a data set of previously collected biometric measurements of WBI to compare with the eddy covariance data from six temperate forests (described in Table 1) within the AmeriFlux network (ameriflux.lbl.gov). We selected temperate forest sites with greater than 10 years of continuous eddy covariance measurements to ensure the time series were sufficiently long to assess the relationship between the two approaches. The sites are mature forests that are representative of mid-to late-successional secondary forest types for each region, including three in the northeastern United States (Howland Forest, US-Ho1; Bartlett Experimental Forest, US-Bar; and Harvard Forest, US-Ha1), two in the midwestern United States (University of Michigan Biological Station, US-UMB; and Morgan Monroe State Forest, US-MMS), and one high elevation forest site in the Rocky Mountain region of the United States (Niwot Ridge, US-NR1). These forests vary in climate, species composition, and standing biomass (Table 1). Throughout the remainder of the manuscript, the sites will be referred to by their AmeriFlux site identifiers (IDs).



#### 2.1.1. Wood Biomass Increment

Annual WBI was estimated for individual trees by reconstructing time series of tree diameter, and then using allometric equations to estimate the change in whole tree biomass over time. Diameter time series was developed from repeated annual measurements at US-Bar (2005–2017), and from increment cores at the remaining five sites, collected between 2012 and 2015 (see Alexander, 2017; Dye et al., 2019; Teets et al., 2018). Trees were selected based on randomly allocated plots located within the eddy covariance tower footprint, with a 10 cm (at breast height) minimum diameter of sampled trees. Data were collected by different researchers at different times, and sampling protocols varied among some of the sites. As a result, our sites had varying plot designs, areas sampled, and number of trees sampled (Table S1 in Supporting Information S1).

To allow for unbiased comparison between measurements among sites, we standardized processing of tree measurement data to annual WBI reported per unit land area. Annual tree-ring measurements were used to sequentially back-calculate tree diameters since tower establishment. First, we estimated inside bark diameters of sampled trees using region-specific and species-specific bark factors (Dixon & Keyser, 2011, 2013; Keyser & Dixon, 2008), then subtracted two times the measured tree-ring width (doubling a radial measurement to calculate diameter change), then used bark factors to calculate the outside bark diameter for each year. Aboveground dry tree biomass was then estimated for each year using genus-specific or family-specific allometric equations (Chojnacky et al., 2014). These equations have the benefit of using consistent equation structures, inputs (all diameter-based equations), outputs (whole-tree aboveground dry-weight biomass), and units (centimeter diameter inputs and kilogram outputs) for all the sites. Belowground biomass was also estimated from Chojnacky et al. (2014) equations and added to aboveground biomass to yield whole-tree dry biomass. These belowground estimates assume consistent root:shoot ratios (approximately 1:5) across sites because we do not have site-level data needed for accurate estimates of belowground biomass. We use these genus-level or family-level equations because regionally developed equations were not available for all species at all sites, and different equation structures can cause bias among species. Whole tree (above-ground and below-ground) biomass increment was then converted to carbon biomass increment by multiplying by the standard carbon fraction of 0.5, which is a reasonable estimate for temperate angiosperm and conifer tree species (Thomas & Martin, 2012). WBI from individual trees were then summed at the plot level and converted to grams of carbon per square meter per year  $(g C m^{-2} yr^{-1}).$ 

Using diameter growth has become a standard approach to estimate changes in tree-level and stand-level biomass accumulation (see Foster et al., 2016). But estimating tree-level biomass increment from diameter growth requires the assumption that whole-tree biomass change is proportional to the change in diameter (Bouriaud et al., 2005). This can introduce uncertainties from annual variability in height growth, root growth, and reproductive effort that are not captured in diameter measurements (Mund et al., 2020). Variability in wood density is another source of uncertainty that affects the accuracy of annual tree-level biomass estimation (Babst et al., 2013) by as much as 30% in some hardwood forests (Delpierre et al., 2016). Additional assumptions are then required when scaling up from tree-level to stand-level estimates of biomass growth. Stand-level WBI estimated from trees does not account for growth of understory plants and saplings. Understory plants account for larger proportions of ecosystem carbon uptake in forests with open canopies compared to dense forests with little light penetration (in this study all of our sites had closed canopies). Predicting WBI back in time also does not account for growth of trees that died prior to sampling (Foster et al., 2014), an assumption that becomes more problematic over long time series. When comparing WBI time series to NEP, carbon sources like decomposition and soil respiration can affect the results because they are largely unrelated to tree carbon balance. Comparisons between WBI and GPP are thought to be more analogous, but unlike NEP, these values are modeled from the data (Desai et al., 2008), and are not measured directly. Despite these several sources of uncertainty, WBI estimated from tree-level measurements is considered a reliable proxy for net primary productivity (Xu et al., 2017).

We also make the assumption that our randomly allocated biometric plots are representative of the footprint of the eddy covariance system. We assessed how tree sampling affected the uncertainty of WBI by conducting Monte Carlo simulations of the sampled trees. For each forest plot, we selected a random sample with replacement of individual trees and assigned them to 1,000 artificial plots with the number of trees based on a Poisson distribution around the average number of trees per plot. For each simulation, we calculated the time series of plot-level WBI by summing the tree-level biomass growth at each yearly timestep. Because we were primarily interested in the uncertainty of the annual variability, and only secondarily interested in the uncertainty in the magnitude



of WBI, we calculated annual uncertainty as the standard deviation of the difference between the yearly WBI and the simulation mean. This is different from the uncertainty in the total magnitude of WBI, because it is the uncertainty in the deviation of each year's measurement to the mean. This uncertainty was estimated to determine how reliably we could detect interannual patterns of WBI. It is important to evaluate the sampling uncertainty because individual trees and species can respond differently to environmental conditions.

#### 2.1.2. Carbon Uptake

At each site, carbon uptake was quantified by measuring  $CO_2$  exchange above the forest canopy using the eddy covariance technique (Baldocchi et al., 1988). The eddy covariance technique measures net ecosystem exchange (NEE), in this case reported in grams of carbon per square meter (g C m<sup>-2</sup> yr<sup>-1</sup>) over a temporal period of interest (in this case months to years). NEE is used to estimate ecosystem respiration (Reco) using an empirical model, and GPP as the difference between NEE and Reco. For comparison to WBI in trees, we report NEP as positive ecosystem uptake (i.e., negative NEE) because this sign convention matches the sign convention of WBI and GPP—carbon moving into the system is positive. NEP can be assumed to equal negative NEE by disregarding sources and sinks for CO<sub>2</sub> not involving conversion to or from organic carbon (Lovett et al., 2006). While some specifications of eddy covariance systems vary among sites (see Table S1 in Supporting Information S1), the differences in measurement approach and instrumentation tend to have a small overall effect on the interannual variability in eddy covariance measurements (Schmidt et al., 2012). Site-level differences in topography, homogeneity, and advection can affect the magnitude of the monthly NEP and GPP estimates. Flux records from these (and other) towers have repeatedly been synthesized to understand the drivers of ecosystem carbon cycling (Biederman et al., 2016; Curtis et al., 2002; Keenan et al., 2014). All datasets are publicly available through the AmeriFlux Network and undergo common quality control, in addition to site-specific quality control and outlier removal. We further quality-checked, filtered, and gap-filled the data as needed. We applied standardized approaches for data quality control to ensure consistent data processing across all sites and years (following those in Richardson et al. [2019]). We filtered measurements with very low horizontal wind speeds ( $<0.5 \text{ m s}^{-1}$ ) and removed outliers with residuals (i.e., between gap-filled and measured values) greater than six times the interquartile range (Richardson & Hollinger, 2007). Next, we used the REddyProc package (Wutzler et al., 2018) in R Statistical Software (R Core Team, 2019) for  $u^*$  (friction velocity) threshold determination, filtering, and gap filling. Gap-filled NEE data sets were then partitioned into GPP and ecosystem respiration using the nighttime partitioning technique (Reichstein et al., 2005), also in the REddyProc package.

We estimated the uncertainty of carbon uptake measurements by adjusting the  $u^*$  threshold used to exclude nocturnal periods that might underestimate nighttime respiration due to advection losses (Barr et al., 2013). The  $u^*$  threshold used to filter nocturnal data is considered one of the largest sources of uncertainty in calculating annual integrals of carbon uptake (Barr et al., 2013). We used single site-specific (i.e., fixed)  $u^*$  thresholds (see Supporting Information) to filter data with insufficient turbulence at the 5%, 50%, and 95% quantiles of the estimated  $u^*$  distributions generated using the REddyProc package. Uncertainty was then estimated as the difference between the mean  $u^*$  threshold (50% quantile) and the upper and lower quantiles. By comparing the annual variability (standard deviation) to the uncertainty, we can test whether the interannual signal exceeded the noise from measurement uncertainty.

#### 2.2. Correlation Matrices

We used correlation matrices to identify the temporal periods of carbon uptake with the strongest correlations to WBI. This has been done previously in studies linking carbon uptake with biometric tree growth (e.g., Babst et al., 2013; Lagergren et al., 2019). Moving windows of carbon uptake integrations can be used to identify relationships that might not be apparent based on calendar-year integrals. Using this approach allows us to connect our results to our competing hypotheses (see Figure 1) and visualize the relationship between numerous periods of carbon uptake WBI simultaneously. All correlations and p values reported are based on Pearson's product moment correlation coefficient. The correlation uncertainties were calculated by taking the standard deviation of 500 Monte Carlo samples with replacement of the WBI measurements and paired integrals.

We also performed a *post hoc* analysis on the relationship between WBI and climate using the same approach as for the carbon uptake analysis. We identified how various seasonal windows of precipitation, air temperature, and incoming short-wave radiation (subsequently referred to as solar radiation) relate to WBI to determine if WBI



is also correlated to the previous year climate conditions. These variables were chosen because they are primary drivers of both WBI and carbon uptake (Zhao & Running, 2010). We used averaged gap-filled mean monthly air temperature and solar radiation collected from the eddy covariance tower, and integrated monthly precipitation from the PRISM interpolated data set (Daly et al., 2008). We also present an analysis comparing annual carbon uptake variables (NEP and GPP) with these same climate variables in the supplemental materials (Figure S2 and S3 in Supporting Information S1).

# 3. Data

The data and R code that support the findings of this study can be found at the repository: Teets (2022; https://doi.org/10.5281/zenodo.6208278). Data used to generate the daily and monthly carbon uptake data are publicly available through AmeriFlux (ameriflux.lbl.gov). The individual DOIs for each AmeriFlux site can be found in Table 1.

# 4. Results

#### 4.1. Magnitude of WBI and Carbon Uptake Measurements

The mature, closed canopy forests selected for our study were net carbon sinks for all the years investigated, with average annual NEP ranging from  $177 \pm 30$  to  $346 \pm 70$  g C m<sup>-2</sup> yr<sup>-1</sup> among sites (Figure 2 and Table 2). The most productive stands (US-MMS and US-Ha1) in terms of both WBI and carbon uptake (NEP and GPP) were deciduous-dominated forests with the largest trees and highest average annual temperatures. The next most productive sites (US-Ho1, US-BAR, and US-UMB) were cold climate, northern forests that had similar mean annual temperatures but varied in species composition from conifer-dominated (US-Ho1) to deciduous-dominated (US-Bar and US-UMB). The least productive forest (US-NR1), a high-elevation conifer stand, had the lowest mean annual temperatures, least annual precipitation, and smallest tree diameters of the forests studied.

Across all sites, the average sampling uncertainty was smaller than the interannual variability for all metrics of forest productivity. The annual variability was generally lower for the two conifer-dominated forests compared to deciduous forests, but this was not always the case. The exception was WBI at US-UMB. Tree plots sampled at US-UMB displayed distinctly different trends in the late 2000s resulting from different species compositions—with one plot increasing, one plot decreasing, and one plot with relatively stable trends in WBI. The resulting average WBI had low interannual variability and high uncertainty. In contrast, the highest interannual variability in carbon uptake was found at US-Ha1, with especially high interannual variability for both NEP and GPP (Figure 2). High interannual variability of NEP at US-Ha1 was most noticeable in the years 2008–2012. It is possible that high levels of soil respiration resulting from increasing temperatures (Finzi et al., 2020) may have caused anomalously low levels of NEP in 2010–2011.

#### 4.2. Relationship Between WBI and Carbon Uptake

We found the average annual WBI was strongly related to the average annual carbon uptake across space (average annual measurements at each site), but had weak correlations across time (based on calendar year sums of carbon uptake). In other words, sites with higher rates of carbon uptake (NEP and GPP) also had higher WBI (Figure 3), but this relationship does not predict the interannual variation in WBI. While the average carbon uptake was a good predictor of average WBI across space (r = 0.95 and 0.79 for NEP and GPP, respectively), individual calendar years of annual carbon uptake had much weaker associations to WBI across sites (r = 0.28 and 0.27 for NEP and GPP, respectively). Long-term means of carbon uptake and WBI are more strongly correlated compared to individual years likely because the averages account for potential lags in allocation (Gough et al., 2008), and because the averages will have lower sampling error compared to individual site years. On average, the magnitude of WBI accounted for the majority of the annual NEP (65%–84%), and 12%–19% of annual GPP. But in individual years, the allocation to WBI was not a simple fraction of the annual carbon uptake (Figures 3c and 3d).

Next, we tested our two competing hypotheses to determine the periods of carbon uptake (from the current and previous year) that were most strongly correlated to WBI at each site (Figure 4). The alternate hypothesis was supported at three of the six sites (US-MMS, US-Ha1, and US-Ho1), indicated by numerous significant positive correlations between WBI and carbon uptake integrals from the previous year. The null hypothesis was supported



🛧 WBI 🔷 NEP 🗗 GPP



**Figure 2.** Annual forest productivity measurements at the six AmeriFlux sites ordered by increasing mean annual temperature. Carbon uptake measurements of NEP (blue) and GPP (red) are presented for the length of overlap with stand-level biometric tree WBI measurements (green) and plot-level WBI (gray). The scales for NEP and tree WBI are represented on the left axis and GPP on the right axis. Error bars for NEP and GPP represent the uncertainty calculated using different  $u^*$  thresholds based on 5%, 50%, and 95% quantiles estimated from 200 bootstrapped iterations. Error bars for WBI represent uncertainty based on a Monte Carlo resampling with replacement of trees. For each of 1,000 simulations, the uncertainty was calculated as the standard deviation of the difference between annual WBI and the mean of that simulation.

at one of the six sites (US-NR1), indicated by the current year's carbon uptake being a better predictor of WBI than carbon uptake from the previous year. At two of the six sites (US-UMB and US-Bar), we found relatively few significant correlations between WBI and carbon uptake, and these sites did not clearly fit either of our two competing hypotheses.

We found that WBI at the two most productive sites (US-MMS and US-Ha1) had significant positive correlations with carbon uptake over much of the current and previous years (Figure 4). This included significant positive correlations between WBI and calendar year integrals of GPP from both the current and previous years. We found that GPP was a slightly better predictor of WBI than NEP at these two sites. The temporal periods of GPP most strongly related to WBI included September of the previous year through current year April at US-Ha1 ( $r = 0.69 \pm 0.10$ , p value < 0.001 for GPP), and previous October through the current year September at US-MMS ( $r = 0.85 \pm 0.08$ , p value <0.001 for GPP). Therefore, we find evidence for a lag in the allocation of assimilates from the previous year at these sites, but carbon uptake during the spring coinciding with growth



# Table 2

Mean Annual Wood Biomass Increment (WBI), Net Ecosystem Productivity (NEP), and Gross Primary Productivity (GPP), and Ecosystem Respiration for Each Site, Listed by Increasing Mean Annual Temperature

Site, (State)	WBI $\pm \sigma_t \pm \sigma_u (gC/m^2)$	GPP $\pm \sigma_t \pm \sigma_u (gC/m^2)$	NEP $\pm \sigma_t \pm \sigma_u (gC/m^2)$	Ecosystem respiration $\pm \sigma_t \pm \sigma_u (gC/m^2)$
US-NR1, (CO)	$146 \pm 15 \pm 4$	$838 \pm 44 \pm 12$	$177 \pm 30 \pm 2$	US-Ha1, (MA)
US-Ho1, (ME)	$161 \pm 17 \pm 6$	$1,331 \pm 90 \pm 14$	$253 \pm 64 \pm 4$	US-MMS, (IN)
US-BAR, (NH)	$168 \pm 34 \pm 14$	$1,303 \pm 86 \pm 66$	$245\pm78\pm2$	$1,058\pm108\pm68$
US-UMB, (MI)	$157 \pm 17 \pm 8$	$1,191 \pm 124 \pm 25$	$198 \pm 72 \pm 3$	$997 \pm 113 \pm 22$
US-Ha1, (MA)	$255 \pm 36 \pm 11$	$1{,}447 \pm 200 \pm 48$	$299 \pm 139 \pm 12$	$1,\!149\pm214\pm59$
US-MMS, (IN)	$295 \pm 47 \pm 19$	$1,526 \pm 91 \pm 28$	$346 \pm 70 \pm 4$	$1,\!180\pm68\pm29$

*Note.* Temporal variability  $(\sigma_t)$  is expressed as the standard deviation of interannual variation in site-level biometric carbon increment and carbon uptake. Uncertainty  $(\sigma_u)$  for biometric measurements is based on Monte Carlo sampling with replacement, and for carbon uptake from different  $u^*$  thresholds listed in Supporting Information.



**Figure 3.** Wood biomass increment (WBI) compared to the annual carbon uptake (net primary productivity in panels (a) and (c) and gross primary productivity in panels (b) and (d)) for each of the six study sites. Panels (a) and (b) represent the relationship between WBI and carbon uptake across space (average annual across all site years), and panels (c) and (d) represent the relationship across time (calendar year totals). Error bars in panels (a) and (b) represent the standard deviation of the yearly totals plotted in the bottom two panels. The slopes were generated using a type II reduced major axis regression. In panels (c) and (d), the average and standard deviation of the slopes and correlations are reported across sites.



Figure 4.



was also significantly associated with growth. Carbon uptake integrals, including both previous fall and current spring, had the best agreement with WBI at these sites.

At the two conifer-dominated sites, we found very divergent relationships between WBI and carbon uptake (Figure 4). The alternate hypothesis was supported at US-Ho1 (WBI was best predicted by the previous year NEP), while the null hypothesis was supported at US-NR1 (WBI was more tightly coupled to NEP from the current year). At US-Ho1, WBI was most strongly correlated with NEP integrals that included the previous summer months. The highest correlations were found with NEP from August of the previous year—a surprisingly strong correlation for just 1 month ( $r = 0.86 \pm 0.19$ , p value < 0.001). In contrast, at US-NR1, WBI had significant positive correlations to NEP from current summer and fall uptake, highest from September to October ( $r = 0.66 \pm 0.19$ ; p value = 0.01). Surprisingly, the strongest relationships at US-NR1 were negative correlations between WBI and January NEP ( $r = -0.75 \pm 0.33$ ; p value = 0.003). Here we show two conifer-dominated forests with similar physiological traits show very different relationships between WBI and carbon uptake.

Compared to the other sites, WBI at US-UMB and US-Bar had fewer temporal periods of carbon uptake with significant correlations to WBI (Figure 4). At US-Bar, the correlations were relatively weak, with the strongest positive correlations found between WBI and NEP integrated from August of the previous year to June of the current year ( $r = 0.61 \pm 0.34$ ; p value = 0.03). At US-UMB, the strongest correlations were found in the winter months, and were positive correlations with GPP from February ( $r = 0.70 \pm 0.32$ ; p value = 0.007). These correlations with winter GPP were surprising because these months have negative NEP and any carbon assimilated is very small and unlikely to contribute meaningfully to WBI. We also note that WBI and US-UMB show significant relationships to winter temperatures from the same period (Figure 5), and we suspect positive correlations between temperatures and GPP in winter are resulting in the spurious relationships we see between WBI and carbon uptake in winter months (see Section 5). Overall, these sites did not clearly support the null or the alternate hypothesis.

We tested whether outliers in monthly NEP or GPP were causing spurious significant relationships with WBI that could affect the interpretation of our findings. We repeated the analysis by replacing any monthly carbon uptake values that were greater than three times the interquartile range different from the monthly median with the median values. We did not find the results and interpretations of this study were affected by extreme values of monthly carbon uptake. For more information, see Figure S1 in Supporting Information S1.

#### 4.3. Occurrence of Lagged Allocation to Biomass Growth

The presence of lagged growth responses to carbon uptake varied widely among the sites used in this study. WBI at two sites (US-MMS and US-Ha1) demonstrated the strongest relationships to integrals including the previous year fall and current year spring. WBI at US-Ho1 was most strongly associated with previous summer carbon uptake. In contrast to all other sites, WBI at US-NR1 demonstrated significant correlations with only current year carbon uptake. We found relatively few significant relationships between WBI and carbon uptake at the two comparatively slow-growing deciduous forests (US-UMB and US-Bar). Our findings indicate that higher productivity sites (higher WBI) showed stronger relationships between WBI and carbon uptake, and those relationships oftentimes extended into the previous year. Less productive (lower WBI) sites had weaker correlations to carbon uptake, and growth was more strongly coupled to current year variables.

#### 4.4. Relationship Between WBI and Climate Variables

Using the same correlation matrix approach as described above, we tested the relationship between WBI and precipitation, air temperature, and solar radiation from the current and previous years to detect lags in different

**Figure 4.** Site level correlation matrices representing the relationship between WBI and net ecosystem productivity (NEP; right panels) gross primary productivity (GPP; right panels). The sites are listed by increasing mean annual temperatures. NEP and GPP are integrated over different temporal periods (start of integral on *x*-axis, and end of integral on *y*-axis). The x- and *y*-axes origins represent the January of year before WBI (the prefix p refers to previous years). The dotted lines separate the two calendar years. The lower left quadrant represents only previous year integrals, the top right quadrant current year integrals, and the top left including portions of both current and previous year metrics. The dashed line represents 12 months integrals (the area above the line represents temporal periods >1 year, and the area below the line is less than 1 year). Shape outlines refer to different seasons: previous year summer ( $\bigcirc$ ), previous year autumn ( $\square$ ), spring ( $\diamondsuit$ ), and summer ( $\bigtriangleup$ ) uptake. Orange colors represent temporal periods with highest positive correlations with tree WBI, and cells with "+" signs represent temporal periods with significant Pearson correlation coefficients (*p* value < 0.05).





Figure 5.



climatological metrics (Figure 5). WBI at each site showed unique responses to climate variables. Sites with strong correlations between WBI and climate were sites that demonstrated relatively weak relationships between WBI and carbon uptake. For example, early season temperatures were more strongly correlated with WBI than the carbon uptake variables at US-UMB ( $r = 0.74 \pm 0.32$ ; p value = 0.003 for February temperatures) and at US-Bar ( $r = 0.78 \pm 0.34$ ; p value = 0.001 for March and April temperatures). Unexpectedly, solar radiation was also negatively correlated to WBI for US-Bar ( $r = -0.79 \pm 0.30$ ; p value = 0.001 for previous April to current February), potentially due to years with higher solar radiation having lower diffuse fraction and precipitation compared to years with low solar radiation. WBI at US-NR1 was most strongly correlated with precipitation from the previous and current growing seasons ( $r = 0.75 \pm 0.25$ ; p value = 0.003 from March of the previous year to current March). Unlike the other sites, WBI at US-NR1, US-UMB, and US-Bar were more strongly associated with climate variables than WBI was with carbon uptake.

We also found significant correlations between WBI and climate variables at the remaining sites (US-Ha1, US-MMS, and US-Ho1), but the relationships were not as strong as those between WBI and carbon uptake. WBI at US-Ha1 exhibited significant correlations to long integrals of precipitation (previous January to current June,  $r = 0.60 \pm 0.13$ , p value = 0.005), and spring temperature (March,  $r = 0.56 \pm 0.16$ , p value = 0.007). At US-MMS, WBI demonstrated significant lagged correlations to previous summer precipitation (previous June-July,  $r = 0.67 \pm 0.14$ , p value = 0.009), and lagged negative correlations with temperature (previous year February to current year April,  $r = -0.69 \pm 0.17$ , p value = 0.006). WBI at US-Ho1 demonstrated relatively weak, but significant, correlations with previous year temperature and precipitation. This analysis demonstrated that WBI in these forests is also frequently affected by lagged growth limitations from climate drivers from the previous year.

We performed a similar analysis comparing annual carbon uptake variables (NEP and GPP) to monthly climate variables in the supplemental materials (Figure S2 and S3 in Supporting Information S1). This analysis shows annual carbon uptake had differing relationships with climate variables than WBI. The climate variables with the highest positive relationships with carbon uptake tended to be from the current year (particularly for GPP), with fewer lagged relationships to climate compared to WBI. This supplemental analysis demonstrates that WBI and carbon uptake are not only affected by different environmental drivers, but the timing of the environmental drivers can differentially affect WBI and carbon uptake.

Lagged growth responses to climate variables were especially common between WBI and precipitation, with all sites demonstrating significant positive correlations between WBI and precipitation from the previous year, except for US-UMB. Current year temperature, however, was more often significantly correlated to WBI than previous year temperature.

#### 4.5. Signal Strength in WBI

We tested our ability to detect relationships between WBI and other variables (i.e., the signal strength) at each site by taking the ratio of interannual variability (the standard deviation of annual WBI) and the uncertainty (estimated using Monte Carlo resampling of individual trees used in the analysis). We use the signal strength similar to a z-score – values greater than 2 (the interannual variability is at least twice as large as the uncertainty) suggests the interannual variability is sufficiently large to confidently detect relationships between WBI and other variables; values less than 1 suggest that the interannual variability is smaller than the uncertainty, and therefore the signal is not distinguishable from the underlying noise. We found the sites with lowest signal strength in WBI had the weakest relationships between WBI and carbon uptake (Figure 6). The two sites (US-Bar, and US-UMB) had a ratio less than two and were both cold-climate, deciduous-dominated forests that exhibited the fewest significant relationships to carbon uptake variables. The two conifer-dominated forests (US-NR1, and US-Ho1) both

**Figure 5.** Site level correlation matrices representing the relationship between WBI and precipitation, temperature, and solar radiation. Sites are listed by increasing mean annual temperature. Variables are integrated for precipitation and averaged for temperature and solar radiation over different temporal periods (start of integral on *x*-axis, and end of integral on *y*-axis). The x- and *y*-axes origins represent the January of year prior to WBI (the prefix p refers to previous years). The dotted lines separate the two calendar years. The lower left quadrant represents only previous year integrals, the top right quadrant current year integrals, and the top left including portions of both current and previous year metrics. The dashed line represents 12 months integrals (the area to above the line represents temporal periods >1 year, and the area below the line is less than 1 year). Shape outlines refer to different seasons: previous year summer ( $\bigcirc$ ), previous year autumn ( $\square$ ), spring ( $\diamondsuit$ ), and summer ( $\bigtriangleup$ ) uptake. Orange colors represent temporal periods with highest positive correlations with tree WBI, and cells with "+" signs represent temporal periods with significant Pearson correlation coefficients (p-value < 0.05).





**Figure 6.** Signal strength of wood biomass increment (WBI) calculated as the standard deviation divided by the uncertainty plotted against the standard deviation of WBI. The horizontal dashed line at a signal strength of 2 (interannual variability was twice as strong as the sampling uncertainty), represents the threshold where relationships between WBI and carbon uptake became apparent.

expressed relatively low interannual variability, but both had relatively high signal strength due to lower tree-to-tree variability. The two most productive forests (US-Ha1 and US-MMS) had comparatively high interannual variability, and sufficient signal strength to demonstrate high correlations between WBI and carbon uptake integrals.

# 5. Discussion

We found strong evidence for lags in carbon allocation to tree growth at three of the six sites based on significant positive correlations between WBI and carbon uptake from the previous year. WBI at the remaining sites did not demonstrate clear relationships with previous year uptake, therefore, we did not find our alternate hypothesis (i.e., WBI is linked to previous year's carbon uptake) to be universally true across these temperate forests. Our study demonstrates that lags in the allocation of stored carbon from previous years to WBI are important mechanisms for explaining WBI, but not in all forest types. In low productivity forests, there was no clear link between WBI and previous year's carbon uptake. Weak relationships between WBI and carbon uptake in the lower productivity forests were likely due to external factors (such as environmental drivers) imposing larger limitations on growth than photosynthetic uptake. Also, the lower interannual variability at these sites makes detecting the relationships between WBI and carbon uptake more difficult, due to the higher relative uncertainty.

Our study supports previous findings that long-term averages of WBI are related to average annual NEP (Pregitzer & Euskirchen, 2004) and GPP (Waring et al., 1998; Zha et al., 2013) across space (i.e., when tested across sites). However, as expected, WBI was less tightly coupled to carbon uptake

across time when using calendar-year timescales (Figure 2). We found that WBI was consistently lower in magnitude than NEP for all sites in this study. This may be partially attributed to the growth of vegetation not included in the analysis (e.g., saplings, understory vegetation, and trees that died during the periods of interest). WBI also does not account for other sinks of carbon in reproductive tissues (Mund et al., 2010), fine root production (Gaudinski et al., 2000) and allocation to microbial symbionts (Schiestl-Aalto et al., 2019). We acknowledge the potential for systematic errors affecting long-term measurements of carbon uptake (Goulden et al., 1996; Hollinger et al., 2004) as well as from scaling up from individual-tree measurements using allometric equations (Alexander et al., 2018). Not accounting for interannual differences in tree ring density can also increase the uncertainty in the interannual variability of WBI (Babst et al., 2013; Delpierre et al., 2016). Despite these several sources of uncertainty, WBI measured in this study accounted for similar proportions of the annual NEP (Anić et al., 2018; Barford et al., 2001) and GPP (Babst et al., 2013; Litton et al., 2007; Luyssaert et al., 2007; Vicca et al., 2012) reported in the literature.

The strongest relationships between WBI and carbon uptake were found at the sites with the highest WBI and highest interannual variability in WBI. Sites with low interannual variability and low signal strength in WBI had relatively weak relationships between growth and uptake. Relationships between WBI and carbon uptake may be more apparent in high-productivity forests because they are known to allocate higher proportions of recently fixed carbon to biomass growth (than to storage) compared to low-productivity forests (Litton et al., 2007; Vicca et al., 2012). Higher growth rates imply fewer environmental growth limitations, and higher demand for growing tissues. Vigorously growing trees have large, high-turnover carbohydrate pools, and growth can be fueled from a combination of current and previous year's assimilates (Carbone et al., 2013). In fertilized forests, trees can rapidly draw down storage from previous years to fuel the periods of increased growth (Goodsman et al., 2010). Here, we find that high productivity forests had growth that was more tightly coupled to carbon uptake from previous years compared to slow-growing forests.

The sites with the strongest positive correlations between WBI and carbon uptake were two deciduous dominated forests with the highest rates (and interannual variability) of WBI, highest annual temperatures, and the largest

trees of the sites studied. These sites experienced favorable conditions for wood growth over the sampling period, based on the increasing trends in carbon uptake and biomass growth in other studies (Dragoni et al., 2011; Finzi et al., 2020; Urbanski et al., 2007). WBI at these two sites (US-Ha1, US-MMS) demonstrated strong correlations to integrals of carbon uptake over much of the current and previous years, indicating WBI was fueled by a combination of stored and current year assimilates. Compared to the other sites in this study, the species compositions at US-Ha1 and US-MMS had relatively high abundances of ring-porous species (85% and 43% of the standing biomass was ring-porous species, respectively, the next highest was found at US-UMB with 25%). Ring-porous species, like those in the genera *Quercus*, initiate xylem growth before leaves develop (Tyree & Cochard, 1996; Wang et al., 1992) – starting growth earlier than diffuse-porous species (D'Orangeville et al., 2021). The large vessels in the earlywood of ring-porous species are prone to embolism in cold temperatures, and therefore the early wood formation is an adaptation to supply water to developing leaves (Zweifel et al., 2006). Therefore, these species must draw on stored nonstructural carbohydrates from previous years to initiate wood growth. An abundance of ring-porous species may be a reason for the correlations between WBI and previous year uptake at these sites.

The relative abundance of conifers versus angiosperms did not explain the differences we found in lagged relationships between WBI and carbon uptake among sites. For example, the two conifer dominated forests (US-Ho1 and US-NR1 - both dominated by Picea species, but with differing codominant genera) had very different relationships between WBI and carbon uptake. We expected these two conifer-dominated forests would have similar lagged relationships between WBI and carbon uptake due to their similar physiology. Compared to angiosperms (diffuse- or ring-porous species), conifers have smaller and more variable stored carbohydrate reserves (Piper & Paula, 2020). Therefore, the coupling between WBI and carbon uptake may depend on the relative amount of stored carbohydrates. An abundance of stored carbohydrates (as found in rapidly growing trees) may allow for higher correspondence between WBI and carbon uptake from the current and previous year (as seen at US-Ho1). Smaller pools of stored carbohydrates will cause growth to be more dependent on current year conditions (as seen at US-NR1). Different environmental conditions at these sites (see Hollinger et al., 1999; Monson et al., 2005) have resulted in differences in stand density (see Table 1) and stand structure. Stand density affects the amount of light penetration into the understory, and more open canopies at US-NR1 could cause carbon uptake to be more influenced by understory vegetation. Differing resource availability might also result in differences in carbon allocation to above- and below-ground resources (Friedlingstein et al., 1999). A more arid climate at US-NR1 likely requires higher carbon allocation to root growth (relative to above ground growth) compared to eastern sites in this study (Ledo et al., 2018).

Perhaps the most prominent example of the alternate hypothesis was shown between WBI and NEP at US-Ho1. Previous work at US-Ho1 had also found WBI responds to previous year's NEP (Richardson et al., 2013; Teets et al., 2018) and these studies were a major motivation for the development of the alternate hypothesis. Our findings here further support the importance of previous year's carbon uptake (and climate variables) for WBI at US-Ho1. We predicted uptake occurring late in the growing season would be strongly associated with WBI in the following year because wood formation in northern forests largely occurs in the spring and early summer (D'Orangeville et al., 2012; Rossi et al., 2006) and ends well before carbon uptake slows down in the winter (Zweifel et al., 2010). Our findings here suggest that uptake from the previous year's summer months was strongly linked to WBI the following year in this evergreen-dominated northern forest, but this does not necessarily translate to other temperate (or conifer-dominated) forests.

The relationships between WBI and carbon uptake at US-Bar were less apparent than other northeastern sites in this study. Similarly, Ouimette et al. (2018) also found that WBI was weakly correlated to current and previous calendar year integrals of uptake at the same site. Our results show WBI exhibited few significant lagged relationships to carbon uptake from the previous summer through current spring, but these relationships were not strong enough to support our alternate hypothesis. WBI was more tightly coupled to current year climate conditions (notably spring temperature). One of the dominant species at US-Bar (*Fagus grandifolia*) is persistently affected by beech bark disease (Leak & Smith, 1996) and has prolific understory resprouting (Pontius et al., 2016). As a result, stand-level growth is negatively affected by relatively high levels of disturbance from pests and tree-to-tree competition compared to other sites. Higher levels of disturbance, and low signal strength in WBI compared to other northeastern sites may cause relatively weak correlations between WBI and carbon uptake at US-Bar.



The site with the lowest signal strength in WBI (US-UMB) did not show significant correlations to any temporal periods with positive NEP, unlike all other sites. This forest showed comparatively little interannual variability in WBI, also noted by Gough et al. (2010), despite changes in the dominant tree species—where early successional species are being replaced by middle-successional and late-successional tree species. We suspect the relatively low interannual variability and high uncertainty in WBI affected our ability to find a clear relationship with carbon uptake. Alternatively, a lack of relationships between WBI and carbon uptake might suggest WBI is limited by factors other than carbon supply (Fatichi et al., 2014). In forests experiencing higher levels of growth limitation, the amount of carbon assimilated can exceed what is needed for growth (Körner, 2003). Despite the low signal strength, WBI at US-UMB was significantly correlated to GPP in January and February, months with very low rates of carbon uptake that can only contribute a minimal quantity of assimilates to WBI. It is likely these significant correlations are due to correlation between winter temperatures and winter GPP. Warm late-winter temperatures in northern forests can hasten snow melt and can lead to earlier (and increased) tree growth (Tardif et al., 2001) and carbon uptake (Keenan et al., 2014; Richardson et al., 2010). Therefore, we infer that the relationships between WBI and late winter GPP at US-UMB are spurious relationships with very little biological justification.

The only high elevation site used in this study (US-NR1) was also the only site to support the null hypothesis that WBI exhibits strong positive correlations to current year, and not previous year, carbon uptake. We found WBI at US-NR1 had significant positive correlations to NEP from the summer and fall corresponding to growth, and showed very little reliance on previous year uptake. WBI was also positively correlated to summer temperatures over the same periods. Warm conditions in the days and nights of summer months are favorable for wood formation in high elevation forests (Körner, 1998). Interestingly, WBI also demonstrated significant negative correlations to winter NEP. This may be explained by the counteracting effects of warm winter temperatures and snow depth in forests that receive most of their precipitation as winter snowfall. Warm conditions at US-NR1 may have positive effects on NEP in the late winter and spring months, but also result in decreased water availability in the summer (Hu et al., 2010; Monson et al., 2005). With sufficient precipitation and warm summer temperatures, we see increases in WBI (this study) and photosynthetic uptake (Monson et al., 2005). Despite significant positive relationships with current year carbon uptake, WBI at US-NR1 was more tightly coupled to recent environmental drivers than to metrics of carbon uptake, and showed no evidence of being limited by carbohydrates carried over from previous years. This is supported by findings that growth is unlikely to be limited by carbohydrates carried over from previous years. This is supported by findings that growth is unlikely to be limited by carbohydrates

# 6. Conclusions

We find the link between WBI and previous year's carbon uptake not only varies among sites, but it largely depends on the rates of wood biomass growth (Friedlingstein et al., 1999) and the external controls on growth. Based on our results, species composition (i.e., the relative abundance of conifers) was not the primary driver of differences found between sites—indicated by the two conifer-dominated forests having very different relationships between WBI and carbon uptake. Instead, the link between WBI and carbon uptake was better explained by forest productivity. Compared to low productivity forests, we found WBI in more productive forests had higher interannual variability (thus higher signal-to-noise ratios), and higher correlations to carbon uptake—with correlations often extending into the previous year. Vigorously growing forests allocate higher proportions of carbon uptake to aboveground biomass compared to slow-growing forests (Hilbert & Reynolds, 1991; Litton et al., 2007; McMurtrie & Dewar, 2013) making the correlation between WBI and carbon uptake more apparent. Stressful environmental conditions can cause the increased allocation to storage as nonstructural carbohydrates (Hoch & Körner, 2012), or fine roots (Kobe et al., 2010) at the expense of structural growth. When structural growth is limited by factors other than photosynthesis (e.g., Lempereur et al., 2015; Mund et al., 2010), WBI can become decoupled from photosynthetic uptake (Muller et al., 2011; Pompa-García et al., 2021).

In summary, our results show evidence that the previous year's carbon uptake is an important driver in explaining annual WBI in some temperate forests. Inconsistent relationships between WBI and carbon uptake in previous studies may have been partially due to lags in allocation of carbon to growth, but other factors such as forest productivity and climate, likely also play important roles. Our findings suggest forest productivity is an important determinant of the use of previously assimilated carbon uptake for tree growth. Forests with strong growth sinks

in WBI can use a combination of carbohydrates from the current year and previously stored assimilates, whereas weak growth sinks are more likely to be limited by current year conditions.

# **Data Availability Statement**

The authors acknowledge the following AmeriFlux sites for their data records: US-NR1, US-Ho1, US-Bar, US-UMB, US-Ha1, and US-MMS. The data and R code that support the findings of this study can be found at the repository: Teets (2022; https://doi.org/10.5281/zenodo.6208278). Data used to generate the daily and monthly carbon uptake data are publicly available through AmeriFlux (ameriflux.lbl.gov). The individual DOIs for each AmeriFlux site can be found in Table 1.

# References

- Alexander, M. R. (2017). Determining the role of stand structure in shaping climate-growth relationships in eastern temperate forests of the US (PhD Dissertation). University of Arizona.
- Alexander, M. R., Rollinson, C. R., Babst, F., Trouet, V., & Moore, D. J. P. (2018). Relative influences of multiple sources of uncertainty on cumulative and incremental tree-ring-derived aboveground biomass estimates. *Trees - Structure and Function*, 32(1), 265–276. https://doi. org/10.1007/s00468-017-1629-0
- Anderegg, W. R. L., Schwalm, C., Biondi, F., Camarero, J. J., Koch, G., Litvak, M., et al. (2015). Pervasive drought legacies in forest ecosystems and their implications for carbon cycle models. *Science*, 349(6247), 528–532. https://doi.org/10.1126/science.aab1833
- Anić, M., Ostrogović Sever, M., Alberti, G., Balenović, I., Paladinić, E., Peressotti, A., et al. (2018). Eddy covariance vs. biometric based estimates of net primary productivity of Pedunculate Oak (*Quercus robur* L.) forest in Croatia during ten years. *Forests*, 9(12), 764. https://doi. org/10.3390/f9120764
- Babst, F., Bouriaud, O., Alexander, R., Trouet, V., & Frank, D. (2014). Toward consistent measurements of carbon accumulation: A multisite assessment of biomass and basal area increment across Europe. *Dendrochronologia*, 32(2), 153–161. https://doi.org/10.1016/j. dendro.2014.01.002
- Babst, F., Bouriaud, O., Papale, D., Gielen, B., Janssens, I. A., Nikinmaa, E., et al. (2013). Above-ground woody carbon sequestration measured from tree rings is coherent with net ecosystem productivity at five eddy-covariance sites. *New Phytologist*, 201(4), 1289–1303. https://doi. org/10.1111/nph.12589
- Baldocchi, D., Hinks, B., & Meyers, T. (1988). Measuring biosphere-atmosphere exchanges of biologically related gases with micrometeorological methods. *Ecology*, 69(5), 1331–1340. https://doi.org/10.2307/1941631
- Barford, C. C., Wofsy, S. C., Goulden, M. L., Munger, J. W., Pyle, E. H., Urbanski, S. P., et al. (2001). Factors controlling long- and short-term sequestration of atmospheric CO2 in a mid-latitude forest. *Science*, 294, 1688–1691. https://doi.org/10.1126/science.1062962
- Barr, A. G., Richardson, A. D., Hollinger, D. Y., Papale, D., Arain, M. A., Black, T. A., et al. (2013). Use of change-point detection for friction-velocity threshold evaluation in eddy-covariance studies. Agricultural and Forest Meteorology, 171–172, 31–45. https://doi.org/10.1016/j. agrformet.2012.11.023
- Biederman, J. A., Scott, R. L., Goulden, M. L., Vargas, R., Litvak, M. E., Kolb, T. E., et al. (2016). Terrestrial carbon balance in a drier world: The effects of water availability in southwestern North America. *Global Change Biology*, 1–13. https://doi.org/10.1111/gcb.13222
- Blanken, P. D., Monson, R. K., Burns, S. P., Bowling, D. R., & Turnipseed, A. A. (1998-). AmeriFlux US-NR1 Niwot Ridge Forest (LTER NWT1). Dataset. https://doi.org/10.17190/AMF/1246088
- Bouriaud, O., Bréda, N., Dupouey, J.-L., & Granier, A. (2005). Is ring width a reliable proxy for stem-biomass increment? A case study in European beech. *Canadian Journal of Forest Research*, 35(12), 2920–2933. https://doi.org/10.1139/x05-202
- Burns, S. P., Blanken, P. D., Turnipseed, A. A., Hu, J., & Monson, R. K. (2015). The influence of warm-season precipitation on the diel cycle of the surface energy balance and carbon dioxide at a Colorado subalpine forest site. *Biogeosciences*, 12(23), 7349–7377. https://doi.org/10.5194/ bg-12-7349-2015
- Carbone, M., Keenan, T., Czimczik, C., Murakami, P., Keefe, J. O., Schaberg, P., et al. (2013). Age, allocation, and availability of nonstructural carbohydrates in red maple. New Phytologist, 200, 1145–1155. https://doi.org/10.1111/nph.12448
- Chojnacky, D. C., Heath, L. S., & Jenkins, J. C. (2014). Updated generalized biomass equations for North American tree species. *Forestry*, 87(1), 129–151. https://doi.org/10.1093/forestry/cpt053
- Clark, D., Brown, S., Kicklighter, D. W., Chambers, J. Q., Thomlinson, J. R., & Ni, J. (2001). Measuring net primary production in forests: Concepts and field methods. *Ecological Applications*, 11(2), 356–370. Retrieved from http://www.esajournals.org/doi/abs/10.1890/1051-076 1(2001)011%255B0356:MNPPIF%255D2.0.CO;2
- Curtis, P. S., Hanson, P. J., Bolstad, P., Barford, C., Randolph, J., Schmid, H., & Wilson, K. B. (2002). Biometric and eddy-covariance based estimates of annual carbon storage in five eastern North American deciduous forests. *Agricultural and Forest Meteorology*, 113(1–4), 3–19. https://doi.org/10.1016/S0168-1923(02)00099-0
- Daly, C., Halbleib, M., Smith, J. I., Gibson, W. P., Doggett, M. K., Taylor, G. H., & Pasteris, P. P. (2008). Physiographically sensitive mapping of climatological temperature and precipitation across the conterminous United States. *International Journal of Climatology*, 28(15), 3–19. https://doi.org/10.1002/joc.1688
- Delpierre, N., Berveiller, D., Granda, E., & Dufrêne, E. (2016). Wood phenology, not carbon input, controls the interannual variability of wood growth in a temperate oak forest. *New Phytologist*, 210, 459–470. https://doi.org/10.1111/nph.13771
- Desai, A. R., Richardson, A. D., Moffat, A. M., Kattge, J., Hollinger, D. Y., Barr, A., et al. (2008). Cross-site evaluation of eddy covariance GPP and RE decomposition techniques. Agricultural and Forest Meteorology, 148(6–7), 821–838. https://doi.org/10.1016/j.agrformet.2007.11.012
- Dietze, M. C., Sala, A., Carbone, M. S., Czimczik, C. I., Mantooth, J. A., Richardson, A. D., & Vargas, R. (2014). Nonstructural carbon in woody plants. Annual Review of Plant Biology, 65, 667–687. https://doi.org/10.1146/annurev-arplant-050213-040054
- Dixon, G. E., & Keyser, C. E. (2011). Northeast (NE) variant overview: Forest vegetation simulator. Fort Collins: U.S. Department of Agriculture, Forest Service, Forest Service Management Service Center.
- Dixon, G. E., & Keyser, C. E. (2013). Central States (CS) variant overview: Forest vegetation simulator. Fort Collins: U.S. Department of Agriculture, Forest Service, Forest Service Management Service Center.

#### Acknowledgments

Funding for AmeriFlux data resources and these core flux sites was provided by the U.S. Department of Energy's Office of Science. AFT would like to thank Anthony J. Garnello for thoughtful feedback to the early stages of the manuscript. The authors acknowledge support from the National Science Foundation (awards DEB 1637685, 1741585, and 1832210). SF acknowledges support from Maine Agricultural and Forest Experiment Station (#ME042118). Research at the Bartlett Experimental Forest is supported by the USDA Forest Service's Northern Research Station. NCAR is sponsored by the National Science Foundation.

- D'Orangeville, L., Itter, M., Kneeshaw, D., Munger, J. W., Richardson, A. D., Dyer, J. M., et al. (2021). Peak radial growth of diffuse-porous species occurs during periods of lower water availability than for ring-porous and coniferous trees. *Tree Physiology*, 1–13. https://doi. org/10.1093/treephys/tpab101
- Dragoni, D., Schmid, H. P., Wayson, C. A., Potter, H., Grimmond, C. S. B., & Randolph, J. C. (2011). Evidence of increased net ecosystem productivity associated with a longer vegetated season in a deciduous forest in south-central Indiana, USA. *Global Change Biology*, 17(2), 886–897. https://doi.org/10.1111/j.1365-2486.2010.02281.x
- Dye, A., Ross Alexander, M., Bishop, D., Druckenbrod, D., Pederson, N., & Hessl, A. (2019). Size–growth asymmetry is not consistently related to productivity across an eastern US temperate forest network. *Oecologia*, 189(2), 515–528. https://doi.org/10.1007/s00442-018-4318-9
- Eglin, T., Francois, C., Michelot, A., Delpierre, N., & Damesin, C. (2010). Linking intra-seasonal variations in climate and tree-ring 13C: A functional modelling approach. *Ecological Modelling*, 221(15), 1779–1797. https://doi.org/10.1016/j.ecolmodel.2010.04.007
- Epron, D., Bahn, M., Derrien, D., Lattanzi, F. A., Pumpanen, J., Gessler, A., et al. (2012). Pulse-labelling trees to study carbon allocation dynamics: A review of methods, current knowledge and future prospects. *Tree Physiology*, 32, 776–798. https://doi.org/10.1093/treephys/tps057
- Estiarte, M., & Penuelas, J. (1999). Excess carbon: The relationship with phenotypical plasticity in storage and defense functions of plants. Orsis, 14, 159–203.
- Fatichi, S., Leuzinger, S., & Körner, C. (2014). Moving beyond photosynthesis: From carbon source to sink-driven vegetation modeling. *New Phytologist*, 201(4), 1086–1095. https://doi.org/10.1111/nph.12614
- Finzi, A. C., Giasson, M., Barker Plotkin, A. A., Aber, J. D., Boose, E. R., Davidson, E. A., et al. (2020). Carbon budget of the Harvard Forest long-term ecological research site: Pattern, process, and response to global change. *Ecological Monographs*, 90(4), e01423. https://doi. org/10.1002/ecm.1423
- Foster, J. R., D'Amato, A. W., & Bradford, J. B. (2014). Looking for age-related growth decline in natural forests: Unexpected biomass patterns from tree rings and simulated mortality. *Oecologia*, 175(1), 363–374. https://doi.org/10.1007/s00442-014-2881-2
- Foster, J. R., Finley, A. O., D'amato, A. W., Bradford, J. B., & Banerjee, S. (2016). Predicting tree biomass growth in the temperate-boreal ecotone: Is tree size, age, competition or climate response most important? *Global Change Biology*, 1–13. https://doi.org/10.1111/gcb.13208

Friedlingstein, P., Joel, G., Field, C. B., & Fung, I. Y. (1999). Toward an allocation scheme for global terrestrial carbon models. *Global Change Biology*, 5(7), 755–770. https://doi.org/10.1046/j.1365-2486.1999.00269.x

- Fritts, H. C. (1976). Tree rings and climate. London, UK: Academic Press.
- Furze, M. E., Huggett, B. A., Aubrecht, D. M., Stolz, C. D., Carbone, M. S., & Richardson, A. D. (2019). Whole-tree nonstructural carbohydrate storage and seasonal dynamics in five temperate species. *New Phytologist*, 221(3), 1466–1477. https://doi.org/10.1111/nph.15462
- Gaudinski, J., Trumbore, S., Davidson, E., & Zheng, S. (2000). Soil carbon cycling in a temperate forest: Radiocarbon-based estimates of residence times, sequestration rates and partitioning of fluxes. *Biogeochemistry*, 51, 33–69. https://doi.org/10.1023/A:1006301010014
- Gea-Izquierdo, G., Bergeron, Y., Huang, J., Lapointe-Garant, M., Grace, J., & Berninger, F. (2014). The relationship between productivity and tree-ring growth in boreal coniferous forests. *Boreal Environment Research*, 19(November), 363–378.
- Goodsman, D. W., Lieffers, V. J., Landhäusser, S. M., & Erbilgin, N. (2010). Fertilization of lodgepole pine trees increased diameter growth but reduced root carbohydrate concentrations. Forest Ecology and Management, 260(10), 1999–1920. https://doi.org/10.1016/j.foreco.2010.08.041
- Gough, C. M., Bohrer, G., & Curtis, P. (1999-). AmeriFlux US-UMB Univ. of Mich. Biological Station. *Dataset*. https://doi.org/10.17190/ AMF/1246107
- Gough, C. M., Flower, C. E., Vogel, C. S., Dragoni, D., & Curtis, P. S. (2009). Whole-ecosystem labile carbon production in a north temperate deciduous forest. Agricultural and Forest Meteorology, 149(9), 1531–1540. https://doi.org/10.1016/j.agrformet.2009.04.006
- Gough, C. M., Hardiman, B. S., Nave, L. E., Bohrer, G., Maurer, K. D., Vogel, C. S., et al. (2013). Sustained carbon uptake and storage following moderate disturbance in a Great Lakes forest. *Ecological Applications*, 23(5), 1202–1215. https://doi.org/10.1890/12-1554.1
- Gough, C. M., Vogel, C. S., Hardiman, B., & Curtis, P. S. (2010). Wood net primary production resilience in an unmanaged forest transitioning from early to middle succession. *Forest Ecology and Management*, 260(1), 36–41. https://doi.org/10.1016/j.foreco.2010.03.027
- Gough, C. M., Vogel, C. S., Schmid, H. P., Su, H.-B., & Curtis, P. S. (2008). Multi-year convergence of biometric and meteorological estimates of forest carbon storage. Agricultural and Forest Meteorology, 148(2), 158–170. https://doi.org/10.1016/j.agrformet.2007.08.004
- Goulden, M. L., Munger, J. W., Fan, S.-M., Daube, B. C., & Wofsy, S. C. (1996). Measurements of carbon storage by long-term eddy correlation: Methods and a critical evaluation of accuracy. *Global Change Biology*, 2, 169–182. https://doi.org/10.1111/j.1365-2486.1996.tb00070.x
- Granier, A., Breda, N., Longdoz, B., Gross, P., & Ngao, J. (2008). Ten years of fluxes and stand growth in a young beech forest at Hesse, North-eastern France. Annals of Forest Science, 64(13), 704. https://doi.org/10.1051/forest:2008052
- Guillemot, J., Martin-Stpaul, N. K., Dufrêne, E., François, C., Soudani, K., Ourcival, J. M., & Delpierre, N. (2015). The dynamic of the annual carbon allocation to wood in European tree species is consistent with a combined source-sink limitation of growth: Implications for modelling. *Biogeosciences*, 12(9), 2773–2790. https://doi.org/10.5194/bg-12-2773-2015
- Hartmann, H., McDowell, N. G., & Trumbore, S. (2015). Allocation to carbon storage pools in Norway spruce saplings under drought and low CO2. *Tree Physiology*, 35(3), 243–252. https://doi.org/10.1093/treephys/tpv019
- Hilbert, D. W., & Reynolds, J. F. (1991). A model allocating growth among leaf proteins, shoot structure, and root biomass to produce balanced activity. Annals of Botany, 68(5), 417–425. https://doi.org/10.1093/oxfordjournals.aob.a088273

Hoch, G., & Körner, C. (1996-). Global patterns of mobile carbon stores in trees at the high-elevation tree line. *Global Ecology and Biogeography*, 21(8), 861–871. https://doi.org/10.1111/j.1466-8238.2011.00731.x

Hollinger, D. Y. (1996-). AmeriFlux US-Ho1 Howland Forest (main tower). Dataset. https://doi.org/10.17190/AMF/1246061

- Hollinger, D. Y., Aber, J., Dail, B., Davidson, E. A., Goltz, S. M., Hughes, H., et al. (2004). Spatial and temporal variability in forest-atmosphere CO2 exchange. *Global Change Biology*, 10, 1689–1706. https://doi.org/10.1111/j.1365-2486.2004.00847.x
- Hollinger, D. Y., Davidson, E. A., Fraver, S., Hughes, H., Lee, J. T., Richardson, A. D., et al. (2021). Multi-decadal carbon cycle measurements indicate resistance to external drivers of change at the Howland Forest AmeriFlux Site. *Journal of Geophysical Research: Biogeosciences*, 126(8), 1–21. https://doi.org/10.1029/2021jg006276
- Hollinger, D. Y., Goltz, S. M., Davidson, E. A., Lee, J. T., Tu, K., & Valentine, H. T. (1999). Seasonal patterns and environmental control of carbon dioxide and water vapour exchange in an ecotonal boreal forest. *Global Change Biology*, 5, 891–902. https://doi. org/10.1046/j.1365-2486.1999.00281.x
- Houghton, R. A., Hall, F., & Goetz, S. J. (2009). Importance of biomass in the global carbon cycle. *Journal of Geophysical Research*, 114(June), 1–13. https://doi.org/10.1029/2009JG000935
- Hu, J., Moore, D. J. P., Burns, S. P., & Monson, R. (2010). Longer growing seasons lead to less carbon sequestration by a subalpine forest. *Global Change Biology*, 16(2), 771–783. https://doi.org/10.1111/j.1365-2486.2009.01967.x
- Kagawa, A., Sugimoto, A., & Maximov, T. C. (2006). Seasonal course of translocation, storage and remobilization of 13C pulse-labeled photoassimilate in naturally growing *Larix gmelinii* saplings. *New Phytologist*, 171(4), 793–804. https://doi.org/10.1111/j.1469-8137.2006.01780.x



- Kannenberg, S. A., Schwalm, C. R., & Anderegg, W. R. L. (2020). Ghosts of the past: How drought legacy effects shape forest functioning and carbon cycling. *Ecology Letters*, 891–901. https://doi.org/10.1111/ele.13485
- Keel, S. G., Siegwolf, R. T. W., & Körner, C. (2006). Canopy CO2 enrichment permits tracing the fate of recently assimilated carbon in a mature deciduous forest. New Phytologist, 172(2), 319–329. https://doi.org/10.1111/j.1469-8137.2006.01831.x
- Keenan, T. F., Gray, J., Friedl, M. A., Toomey, M., Bohrer, G., Hollinger, D. Y., et al. (2014). Net carbon uptake has increased through warming-induced changes in temperate forest phenology. *Nature Climate Change*, 4, 598–604. https://doi.org/10.1038/NCLIMATE2253
- Keyser, C. E., & Dixon, G. E. (2008). Central Rockies (CR) variant overview: Forest vegetation simulator. Fort Collins: U.S. Department of Agriculture, Forest Service, Forest Service Management Service Center
- Kobe, R. K., Iyer, M., & Walters, M. B. (2010). Optimal partitioning theory revisited: Nonstructural carbohydrates dominate root mass responses to nitrogen. *Ecology*, 91(1), 166–179. https://doi.org/10.1890/09-0027.1
- Körner, C. (1998). A re-assessment of high elevation treeline positions and their explanation. *Oecologia*, 115(4), 445–459. https://doi.org/10.1007/s004420050540
- Körner, C. (2003). Carbon limitation in trees. Journal of Ecology, 91(1), 4–17. https://doi.org/10.1046/j.1365-2745.2003.00742.x
- Kozlowski, T. T. (1992). Carbohydrate sources and sinks in woody plants. The Botanical Review, 58(2), 107-222. https://doi.org/10.1007/ bf02858600
- Kozlowski, T. T., & Pallardy, S. G. (2002). Acclimation and adaptive responses of woody plants to environmental stresses. *The Botanical Review*, 68(2), 270–334. https://doi.org/10.1663/0006-8101(2002)068[0270:AAAROW]2.0.CO;2
- Kuptz, D., Fleischmann, F., Matyssek, R., & Grams, T. E. E. (2011). Seasonal patterns of carbon allocation to respiratory pools in 60-yr-old deciduous (*Fagus sylvatica*) and evergreen (*Picea abies*) trees assessed via whole-tree stable carbon isotope labeling. *New Phytologist*, 191, 160–172. https://doi.org/10.1111/j.1469-8137.2011.03676.x
- Lagergren, F., Jönsson, A. M., Linderson, H., & Lindroth, A. (2019). Time shift between net and gross CO2 uptake and growth derived from tree rings in pine and spruce. *Trees - Structure and Function*, 33(3), 765–776. https://doi.org/10.1007/s00468-019-01814-9
- Leak, W. B., & Smith, M. L. (1996). Sixty years of management and natural disturbance in a New England forested landscape. Forest Ecology and Management, 81(1–3), 63–73. https://doi.org/10.1016/0378-1127(95)03662-8
- Ledo, A., Paul, K. I., Burslem, D. F. R. P., Ewel, J. J., Barton, C., Battaglia, M., et al. (2018). Tree size and climatic water deficit control root to shoot ratio in individual trees globally. *New Phytologist*, 217(1), 8–11. https://doi.org/10.1111/nph.14863
- Lempereur, M., Martin-Stpaul, N. K., Damesin, C., Joffre, R., Ourcival, J. M., Rocheteau, A., & Rambal, S. (2015). Growth duration is a better predictor of stem increment than carbon supply in a Mediterranean oak forest: Implications for assessing forest productivity under climate change. New Phytologist, 207(3), 579–590. https://doi.org/10.1111/nph.13400
- Litton, C. M., Raich, J. W., & Ryan, M. G. (2007). Carbon allocation in forest ecosystems. Global Change Biology, 13(10), 2089–2109. https:// doi.org/10.1111/j.1365-2486.2007.01420.x
- Lovett, G. M., Cole, J. J., & Pace, M. L. (2006). Is net ecosystem production equal to ecosystem carbon accumulation? *Ecosystems*, 9(1), 152–155. https://doi.org/10.1007/s10021-005-0036-3
- Luyssaert, S., Inglima, I., Jung, M., Richardson, A. D., Reichstein, M., Papale, D., et al. (2007). CO2 balance of boreal, temperate, and tropical forests derived from a global database. *Global Change Biology*, 13(12), 2509–2537. https://doi.org/10.1111/j.1365-2486.2007.01439.x
- McMurtrie, R. E., & Dewar, R. C. (2013). New insights into carbon allocation by trees from the hypothesis that annual wood production is maximized. *New Phytologist*, *199*(4), 981–990. https://doi.org/10.1111/nph.12344
- Mildner, M., Bader, M. K. F., Leuzinger, S., Siegwolf, R. T. W., & Körner, C. (2014). Long-term 13C labeling provides evidence for temporal and spatial carbon allocation patterns in mature *Picea abies*. *Oecologia*, *175*(3), 747–762. https://doi.org/10.1007/s00442-014-2935-5
- Monson, R. K., Sparks, J. P., Rosenstiel, T. N., Scott-Denton, L. E., Huxman, T. E., Harley, P. C., et al. (2005). Climatic influences on net ecosystem CO2 exchange during the transition from wintertime carbon source to springtime carbon sink in a high-elevation, subalpine forest. *Oecologia*, 146(1), 130–147. https://doi.org/10.1007/s00442-005-0169-2
- Muller, B., Pantin, F., Génard, M., Turc, O., Freixes, S., Piques, M., & Gibon, Y. (2011). Water deficits uncouple growth from photosynthesis, increase C content, and modify the relationships between C and growth in sink organs. *Journal of Experimental Botany*, 62(6), 1715–1729. https://doi.org/10.1093/jxb/erq438
- Mund, M., Herbst, M., Knohl, A., Matthäus, B., Schumacher, J., Schall, P., et al. (2020). It is not just a 'trade-off': Indications for sink- and source-limitation to vegetative and regenerative growth in an old-growth beech forest. New Phytologist, 226(1), 111–125. https://doi. org/10.1111/nph.16408
- Mund, M., Kutsch, W. L., Wirth, C., Kahl, T., Knohl, A., Skomarkova, M. V., & Schulze, E.-D. (1991-). The influence of climate and fructification on the inter-annual variability of stem growth and net primary productivity in an old-growth, mixed beech forest. *Tree Physiology*, 30(6), 689–704. https://doi.org/10.1093/treephys/tpq027
- Munger, J. W. (1991-). AmeriFlux US-Ha1 Harvard Forest EMS tower (HFR1). Dataset. https://doi.org/10.17190/AMF/1246059
- Novick, K., & Phillips, R. (1999-). AmeriFlux US-MMS Morgan Monroe State Forest. Dataset. https://doi.org/10.17190/AMF/1246080
- Ogle, K., Barber, J. J., Barron-Gafford, G. A., Bentley, L. P., Young, J. M., Huxman, T. E., et al. (2015). Quantifying ecological memory in plant and ecosystem processes. *Ecology Letters*, 18(3), 221–235. https://doi.org/10.1111/ele.12399
- Ohtsuka, T., Saigusa, N., & Koizumi, H. (2009). On linking multiyear biometric measurements of tree growth with eddy covariance-based net ecosystem production. *Global Change Biology*, 15(4), 1015–1024. https://doi.org/10.1111/j.1365-2486.2008.01800.x
- Ouimette, A. P., Ollinger, S. V., Richardson, A. D., Hollinger, D. Y., Keenan, T. F., Lepine, L. C., & Vadeboncoeur, M. A. (2018). Carbon fluxes and interannual drivers in a temperate forest ecosystem assessed through comparison of top-down and bottom-up approaches. Agricultural and Forest Meteorology, 256–257, 420–430. https://doi.org/10.1016/j.agrformet.2018.03.017
- Pan, Y., Birdsey, R., Fang, J., Houghton, R., Kauppi, P., Kurz, W., et al. (2011). A large and persistent carbon sink in the world's forests. *Science*, 333, 988–994. http://www.sciencemag.org/content/333/6045/988.short
- Piper, F. I., & Paula, S. (2020). The role of nonstructural carbohydrates storage in forest resilience under climate change. Current Forestry Reports, 6(1), 1–13. https://doi.org/10.1007/s40725-019-00109-z
- Pompa-García, M., Camarero, J. J., Colangelo, M., & Gallardo-Salazar, J. L. (2021). Xylogenesis is uncoupled from forest productivity. Trees -Structure and Function, 35(4), 1123–1134. https://doi.org/10.1007/s00468-021-02102-1
- Pontius, J., Halman, J. M., & Schaberg, P. G. (2016). Seventy years of forest growth and community dynamics in an undisturbed northern hardwood forest. *Canadian Journal of Forest Research*, 46(7), 959–967. https://doi.org/10.1139/cjfr-2015-0304
- Pregitzer, K. S., & Euskirchen, E. S. (2004). Carbon cycling and storage in world forests: Biome patterns related to forest age. *Global Change Biology*, 10(12), 2052–2077. https://doi.org/10.1111/j.1365-2486.2004.00866.x
- R Core Team. (2019). R: A language and environment for statistical computing. Vienna, Austria: R Foundation for Statistical Computing. Retrieved from http://www.R-project.org/



Reichstein, M., Falge, E., Baldocchi, D., Papale, D., Aubinet, M., Berbigier, P., et al. (2004-). On the separation of net ecosystem exchange into assimilation and ecosystem respiration: Review and improved algorithm. *Global Change Biology*, 11(9), 1424–1439. https://doi. org/10.1111/j.1365-2486.2005.001002.x

Richardson, A. D. (2004-). AmeriFlux US-Bar Bartlett Experimental Forest. Dataset. https://doi.org/10.17190/AMF/1246030

- Richardson, A. D., Black, T. A., Ciais, P., Delbart, N., Friedl, M. A., Gobron, N., et al. (2010). Influence of spring and autumn phenological transitions on forest ecosystem productivity. *Philosophical Transactions of the Royal Society B: Biological Sciences*, 365(1555), 3227–3246. https://doi.org/10.1098/rstb.2010.0102
- Richardson, A. D., Carbone, M. S., Keenan, T. F., Czimczik, C. I., Hollinger, D. Y., Murakami, P., et al. (2013). Seasonal dynamics and age of stemwood nonstructural carbohydrates in temperate forest trees. New Phytologist, 197(3), 850–861. https://doi.org/10.1111/nph.12042
- Richardson, A. D., & Hollinger, D. Y. (2007). A method to estimate the additional uncertainty in gap-filled NEE resulting from long gaps in the CO2 flux record. Agricultural and Forest Meteorology, 147(3–4), 199–208. https://doi.org/10.1016/j.agrformet.2007.06.004
- Richardson, A. D., Hollinger, D. Y., Shoemaker, J. K., Hughes, H., Savage, K., & Davidson, E. A. (2019). Six years of ecosystem-atmosphere greenhouse gas fluxes measured in a sub-boreal forest. *Scientific Data*, 6(1), 117. https://doi.org/10.1038/s41597-019-0119-1
- Rocha, A. V., Goulden, M. L., Dunn, A. L., & Wofsy, S. C. (2006). On linking interannual tree ring variability with observations of whole-forest CO2 flux. *Global Change Biology*, 12(8), 1378–1389. https://doi.org/10.1111/j.1365-2486.2006.01179.x
- Rossi, S., Deslauriers, A., Anfodillo, T., Morin, H., Saracino, A., Motta, R., et al. (2006). Conifers in cold environments synchronize maximum growth rate of tree-ring formation with day length. *New Phytologist*, *170*, 301–310. https://doi.org/10.1111/j.1469-8137.2006.01660.x
- Russell, M. B., Woodall, C. W., Fraver, S., D'Amato, A. W., Domke, G. M., & Skog, K. E. (2014). Residence times and decay rates of downed woody debris biomass/carbon in eastern US forests. *Ecosystems*, 17(5), 765–777. https://doi.org/10.1007/s10021-014-9757-5
- Sala, A., Woodruff, D. R., & Meinzer, F. C. (2012). Carbon dynamics in trees: Feast or famine? *Tree Physiology*, 32(6), 764–775. https://doi.org/10.1093/treephys/tpr143
- Schiestl-Aalto, P., Ryhti, K., Mäkelä, A., Peltoniemi, M., Bäck, J., & Kulmala, L. (2019). Analysis of the NSC storage dynamics in tree organs reveals the allocation to belowground symbionts in the framework of whole tree carbon balance. *Frontiers in Forests and Global Change*, 2(May), 1–14. https://doi.org/10.3389/ffgc.2019.00017
- Schmid, H. P., Grimmond, C. S. B., Cropley, F., Offerle, B., & Su, H. B. (2000). Measurements of CO<sub>2</sub> and energy fluxes over a mixed hardwood forest in the mid-western United States. Agricultural and Forest Meteorology, 103(4), 357–374. https://doi.org/10.1016/S0168-1923(00)00140-4
- Schmidt, A., Hanson, C., Stephen Chan, W., & Law, B. E. (2012). Empirical assessment of uncertainties of meteorological parameters and turbulent fluxes in the AmeriFlux network. *Journal of Geophysical Research G: Biogeosciences*, 117(4). https://doi.org/10.1029/2012JG002100
- Tardif, J., Brisson, J., & Bergeron, Y. (2001). Dendroclimatic analysis of Acer saccharum, Fagus grandifolia, and Tsuga canadensis from an old-growth forest, southwestern Quebec. Canadian Journal of Forest Research, 31(9), 1491–1501. https://doi.org/10.1139/cjfr-31-9-1491 Teets, A. (2022). AmeriFlux Wood biomass increment (v1.2). Zenodo, https://doi.org/10.5281/zenodo.6208278
- Teets, A., Fraver, S., Hollinger, D. Y., Weiskittel, A. R., Seymour, R. S., & Richardson, A. D. (2018). Linking annual tree growth with eddy-flux measures of net ecosystem productivity across twenty years of observation in a mixed conifer forest. Agricultural and Forest Meteorology, 249(February), 479–487. https://doi.org/10.1016/j.agrformet.2017.08.007
- Thomas, S. C., & Martin, A. R. (2012). Carbon content of tree tissues: A synthesis. Forests, 3(2), 332–352. https://doi.org/10.3390/f3020332
- Turnipseed, A. A., Blanken, P. D., Anderson, D. E., & Monson, R. K. (2002). Energy budget above a high-elevation subalpine forest in complex topography. Agricultural and Forest Meteorology, 110(3), 177–201. https://doi.org/10.1016/S0168-1923(01)00290-8
- Tyree, M., & Cochard, H. (1996). Summer and winter embolism in oak: Impact on water relations. Annals of Forest Science, 53, 173–180. https://doi.org/10.1051/forest:19960201
- Urbanski, S., Barford, C., Wofsy, S., Kucharik, C., Pyle, E., Budney, J., et al. (2007). Factors controlling CO2 exchange on timescales from hourly to decadal at Harvard Forest. *Journal of Geophysical Research: Biogeosciences*, 112(2), 1–25. https://doi.org/10.1029/2006JG000293
- Vicca, S., Luyssaert, S., Peñuelas, J., Campioli, M., Chapin, F. S., Ciais, P., et al. (2012). Fertile forests produce biomass more efficiently. *Ecology Letters*, 15(6), 520–526. https://doi.org/10.1111/j.1461-0248.2012.01775.x
- Wang, J., Ives, N. E., & Lechowiczt, M. J. (1992). The relation of foliar phenology to xylem embolism in trees. Functional Ecology, 6(4), 469–475. https://doi.org/10.2307/2389285
- Waring, R. H., Landsberg, J. J., & Williams, M. (1998). Net primary production of forests: A constant fraction of gross primary production? *Tree Physiology*, 18(2), 129–134. https://doi.org/10.1093/treephys/18.2.129
- Wutzler, T., Lucas-Moffat, A., Migliavacca, M., Knauer, J., Sickel, K., Šigut, L., et al. (2018). Basic and extensible post-processing of eddy covariance flux data with REddyProc. *Biogeosciences*, 15(16), 5015–5030. https://doi.org/10.5194/bg-15-5015-2018
- Xu, K., Wang, X., Liang, P., An, H., Sun, H., Han, W., & Li, Q. (2017). Tree-ring widths are good proxies of annual variation in forest productivity in temperate forests. *Scientific Reports*, 7(1), 1–8. https://doi.org/10.1038/s41598-017-02022-6
- Zha, T. S., Barr, A. G., Bernier, P. Y., Lavigne, M. B., Trofymow, J. A., Amiro, B. D., et al. (2013). Gross and aboveground net primary production at Canadian forest carbon flux sites. Agricultural and Forest Meteorology, 174–175, 54–64. https://doi.org/10.1016/j.agrformet.2013.02.004
- Zhao, M., & Running, S. W. (2010). Drought-induced reduction in global terrestrial net primary production from 2000 through 2009. *Science*, *329*(5994), 940–943. https://doi.org/10.1126/science.1192666
- Zweifel, R., Eugster, W., Etzold, S., Dobbertin, M., Buchmann, N., & Häsler, R. (2010). Link between continuous stem radius changes and net ecosystem productivity of a subalpine Norway spruce forest in the Swiss Alps. *New Phytologist, 187*(3), 819–830. https://doi.org/10.1111/j.1469-8137.2010.03301.x
- Zweifel, R., & Sterck, F. (2018). A conceptual tree model explaining legacy effects on stem growth. *Frontiers in Forests and Global Change*, *1*(November), 1–9. https://doi.org/10.3389/ffgc.2018.00009
- Zweifel, R., Zimmermann, L., Zeugin, F., & Newbery, D. M. (2006). Intra-annual radial growth and water relations of trees : Implications towards a growth mechanism. *Journal of Experimental Botany*, 57(6), 1445–1459. https://doi.org/10.1093/jxb/erj125